Author's response to reviews

Title: Acquired heterotopic ossification in hips and knees following encephalitis: case report and literature review

Authors:

Xianghong Zhang (zhangxianghong2008@126.com)
Shuo Jie (csxhs@126.com)
Tang Liu (liutang1981@126.com)
Xiangsheng Zhang (xiangshengzhang@hotmail.com)

Version: 4 Date: 13 February 2014

Author's response to reviews: see over
Acquired heterotopic ossification in hips and knees following encephalitis: case report and literature review

*Xianghong Zhang, MD, Orthopaedic Surgeon;

*Shuo Jie, MD, Orthopaedic Surgeon;

Tang Liu, MD, Orthopaedic Surgeon;

Xiangsheng Zhang, MD, Orthopaedic Surgeon, Professor.

*Xianghong Zhang & *Shuo Jie are co-first authors.

Correspondence author: Tang Liu.

Address: Dept of Orthopedics, the Second Xiangya Hospital, Central South University; Changsha, Hunan, P.R. China. (410011).

E-mail: Xianghong Zhang: zhangxianghong2008@126.com

Tang Liu: liutang1981@126.com

Shuo Jie: csxhs@126.com

Xiangsheng Zhang: xiangshengzhang@hotmail.com

Tel: 86-7315295127